

Funding and conclusions of network meta-analyses on targeted therapies in inflammatory diseases: an overview

Robin Guelimi, Sivem Afach, Thomas Bettuzzi, Antoine Meyer, Guillaume Padern, Zenas Yiu, Florian Naudet, Émilie Sbidian, Laurence Le Cleach

▶ To cite this version:

Robin Guelimi, Sivem Afach, Thomas Bettuzzi, Antoine Meyer, Guillaume Padern, et al.. Funding and conclusions of network meta-analyses on targeted therapies in inflammatory diseases: an overview. Journal of Clinical Epidemiology, 2024, 172, pp.111411. 10.1016/j.jclinepi.2024.111411. hal-04652743

HAL Id: hal-04652743 https://hal.science/hal-04652743

Submitted on 18 Jul2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License





Journal of Clinical Epidemiology 172 (2024) 111411

Funding and conclusions of network meta-analyses on targeted therapies in inflammatory diseases: an overview

Robin Guelimi^{a,b,*}, Sivem Afach^a, Thomas Bettuzzi^{a,b}, Antoine Meyer^c, Guillaume Padern^d, Zenas Yiu^{e,f}, Florian Naudet^{g,h}, Emilie Sbidian^{a,b}, Laurence Le-Cleach^{a,b}

^aEpiDermE EA 7379, Université Paris Est Créteil, Créteil, F-94010, France

^bDepartment of Dermatology, Assistance Publique-Hôpitaux de Paris, Henri Mondor Hospital, Créteil, F-94010, France

^cDepartment of Gastroenterology, University Hospital of Bicêtre, Assistance Publique-Hôpitaux de Paris and Université Paris-Saclay,

Le Kremlin Bicêtre, France

^dIRMB, University of Montpellier, Inserm U1183, CHU Montpellier, Montpellier, France

^eFaculty of Biology, Medicine and Health, Division of Musculoskeletal and Dermatological Sciences, School of Biological Sciences, The University of

Manchester, Manchester, United Kingdom

^fSalford Royal Hospital, Northern Care Alliance NHS Foundation Trust, Manchester Academic Health Science Centre,

Manchester, M6 8HD, United Kingdom

^gUniv Rennes, CHU Rennes, Inserm, Irset (Institut de Recherche en Santé, Environnement et Travail) - UMR_S 1085, CIC 1414

[(Centre d'Investigation Clinique de Rennes)], F-35000, Rennes, France

^hInstitut Universitaire de France (IUF), Paris, France

Accepted 3 June 2024; Published online 7 June 2024

Abstract

Objectives: To explore the association between industry funding and network meta-analyses' (NMAs) conclusion, and the use in Clinical Practice Guidelines (CPGs) of NMAs.

Study Design and Setting: This was an overview of NMAs and CPGs. We searched PubMed/MEDLINE, Epistemonikos, and several guideline databases up to February 18th 2023. We included CPGs from the last 5 years and NMAs of randomized controlled trials that evaluated targeted therapies in immune-mediated inflammatory diseases. Data extraction and outcome assessments were done in duplicate by independent authors.

Results: We included 216 NMAs and 99 CPGs. 31% (67/216) were industry-funded. The proportion of industry-funded NMAs that cited one treatment as being best was 44% (25/57) compared to 26% (30/116) for nonindustry-funded (OR = 2.24 [1.15-4.39]; aOR = 1.76 [0.81-3.81]). The abstract's conclusion of 39/67 (58%) industry-funded and 69/149 (46%) nonindustry-funded NMAs were considered unsupported by the results (OR = 1.61 [0.90-2.89]; aOR = 1.40 [0.71-2.78]). All industry-funded NMAs that cited one treatment as best cited their own sponsored drug. 59/99 (60%) CPGs included at least one NMA, with 23/59 (39%) of them citing industry-funded NMAs.

Conclusions: We did not find evidence that industry-funded NMAs were more likely to have unsupported conclusions or to cite only one treatment as being best in their conclusions compared to non-industry-funded NMAs. However, almost all industry-funded NMAs favored their own treatments. Even though 40% of the CPGs did not rely on NMA, over a third of those who did used industry-funded NMAs. Limitations include the possible misclassification due to undisclosed funding and potential confounders that have not been accounted for. © 2024 The Author(s). Published by Elsevier Inc. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Keywords: Network meta-analysis; Funding; Targeted therapies; Systematic review; Immune-mediated inflammatory diseases; Coherence

osf.io/wgydt) and AIMOS top-up scholarship. Neither of the two sources of funding was involved in carrying out this study. * Corresponding author. EpiDermE, Université Paris Est Créteil, 61

Avenue du Général de Gaulle, 94010, Créteil, France. *E-mail address:* robin.guelimi@gmail.com (R. Guelimi).

https://doi.org/10.1016/j.jclinepi.2024.111411

0895-4356/© 2024 The Author(s). Published by Elsevier Inc. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Protocol availability: OSF Registration number osf.io/wgydt (2023-02-18).

Funding: This work was Assistance Publique des Hôpitaux de Paris (AP-HP) research grant « Concours de la médaille » (registrationr

Plain language summary

Network meta-analyses (NMAs) are a type of study that allows to compare in a single analysis many different drugs and can provide a ranking of these drugs from the most desirable to the least desirable (e.g.: from most to least effective). This type of study has become very popular to guide clinical decisions and to help during the writing of practice guidelines. However, some concerns have emerged regarding the potential influence of industry funding on the results and conclusions of published NMAs.

The objective of our study was to assess if there was an influence of funding on the conclusions of the NMAs evaluating innovative drugs in chronic inflammatory diseases. We also evaluated if practice guidelines relied on these studies and if they were industry-funded or not.

Our study shows that industry-funded NMAs largely promoted their own treatment. Yet, even though we had some concerns regarding the relevance or quality of industry-funded studies, we found some flaws in both industry-funded and nonindustry-funded studies. The use of NMAs was modest among guidelines, but a third of those that relied on them cited industry-funded NMAs, which recommended their own treatment.

Our study suggests that efforts are still needed to improve the interpretation and the accurate presentation of NMAs results, both in industry and nonindustry-funded studies. When interpreting the results of a network meta-analysis, bias related to industry funding should be taken into consideration.

1. Introduction

The autoimmune market has been evaluated to increase from \$154 billion worldwide in 2021 up to \$193 billion over the next decade [1], with one of these growth key drivers being the rapid development of biologics and small molecules (eg, JAK inhibitors) to treat chronic immune-mediated inflammatory diseases (IMIDs) [2]. These diseases, including in particular psoriasis, ankylosing spondylitis, rheumatoid arthritis, inflammatory bowel diseases (IBDs), and psoriatic arthritis, constitute a clinically diverse group of conditions with common underlying pathogenetic features [3]. Besides concerns from pharmaceutical companies over economic competitiveness, reliable knowledge on the relative efficacy and safety of these treatments is crucial for clinical practice guidelines (CPGs) developers and clinicians [4-6]. However, most conducted randomized controlled trials used placebo as comparator despite the availability of active treatments [7,8]. To that end, network meta-analysis (NMA) has been useful in addressing some of these gaps in comparative effectiveness research [9].

Concerns have been raised regarding the increasing number of redundant and discordant NMAs being published [10,11]. A scoping review showed that among a sample of 658 NMAs published between 2013 and 2018, industry-funded NMAs were more likely to suggest favorable conclusions to the sponsor product, with 92% of the 136 included industry-funded NMAs recommending their company's intervention [12]. In a previous study, we showed that industry-funded NMAs on the systemic treatments of psoriasis almost systematically cited their treatment as being best [13].

The objective of our study was to evaluate the effect of funding on the conclusions and the use in CPGs of the NMAs' evaluating the efficacy or safety of the authorized targeted therapies used in a group of chronic IMIDs.

2. Methods

We conducted an overview of the NMAs that evaluated the authorized targeted therapies used in several chronic IMIDs of interest as well as an overview of the CPGs that made recommendations on their management using targeted therapies. Both were based on a protocol registered prior to data collection on the Open Science Framework on February the 18th 2023 (osf.io/wgydt). Reporting was performed according to the Preferred Reporting Items for Overviews of Reviews statement (Supplementary Table 17) [14].

2.1. Eligibility criteria

2.1.1. Overview of the NMAs

We considered NMAs of randomized controlled trials that included adults with moderate-to-severe IBD, active ankylosing spondylitis, active psoriatic arthritis, moderate-to-severe psoriasis, or active rheumatoid arthritis, and that evaluated the efficacy or safety of the authorized targeted therapies (Supplementary Table 1) compared to placebo or with an active comparator. We excluded from our overview cost-effectiveness studies, NMAs that only compared biosimilars with their respective originators, and NMAs that used individual patient data (IPD). NMAs that only compared different dosages or regimens of the same intervention were also excluded.

2.1.2. Overview of the CPGs

We carried out a systematic review to include any CPGs published during the last 5 years (i.e., January 2018) concerning adult patients and covering the management with targeted therapies of the IMIDs of interest, developed by local, regional, national, or international groups or affiliated

What is new?

Key findings

- Industry-funded network meta-analyses (NMAs) largely promoted their own treatment.
- Issues regarding relevance and quality were present in the included studies, irrespective of the source of funding.
- The use of NMA was modest among clinical practice guidelines (CPGs).

What this adds to what was known?

- We assessed the NMAs' abstracts to evaluate if their conclusions were supported by the reported results.
- Source of funding aside, half of the NMAs' conclusions were not supported by the reported results.
- A third of the CPGs that cited NMAs cited industry-funded NMAs, which recommended their own treatment.

What is the implication and what should change now?

- Efforts are still needed to improve the interpretation and the accurate reporting of NMAs' results, both in industry- and nonindustry- funded studies.
- Bias related to funding should be accounted for when appraising the results of a network meta-analysis (NMA).

governmental organizations. Guidelines or consensus statements not informed by systematic review or explicit assessment regarding the body of evidence, editorials and commentaries, consensus based on the opinion of expert panels, guidelines targeting policymakers (eg, health technology assessments) were excluded.

2.2. Study identification

We aimed to identify all relevant NMAs, regardless of language. We searched from inception in PubMed/MEDLINE, Epistemonikos, and International Prospective Register of Systematic Reviews. For psoriasis, we updated our previous literature search from February 2021 [13]. The reference lists of the included studies were also screened.

To identify the guidelines of interest, we searched starting in January 2018, Pubmed/MEDLINE and several guidelines depositories (Supplementary Table 3). We also screened the reference list of the included guidelines and the reference list provided by the Global Guidelines in Dermatology Mapping Project for psoriasis [15]. When multiple versions of the same CPG were identified, the most updated CPG was considered.

The selection process was conducted through Covidence (https://www.covidence.org/) up to February 18, 2023, by the review authors (L.L.C., R.G., T.B., A.M., and G.P.). After excluding duplicates, a single author (R.G.) screened every title and abstract. Independent authors (L.L.C., R.G., A.M., and G.P.) then assessed in duplicate the remaining full texts. In case of missing full text, we contacted studies' authors. Discrepancies were solved through discussion. Full search equations can be found in the Supplementary Tables 2 and 3.

2.3. Data extraction

Data extraction was performed in duplicate by independent authors (L.L.C., R.G., and T.B.) following a form previously tested. We extracted the main characteristics of the NMAs, characteristics of the studied population, included interventions, outcomes, number of included studies, of interventions, and of interventions cited as being best in the abstract's conclusion. The study exposure groups were considered industry-funded, if the source of funding was a pharmaceutical company, or nonindustry-funded if the source of funding was either public or academic, if there was no funding or if the source of funding was not reported. In the case of cofunding from different categories (eg, public and industry funding), the study was classified in the category "industry-funded". Authors' financial conflicts of interest (COI) were extracted from the authors' affiliation section and from the publication's COI disclosure. Any types of financial ties with pharmaceutical companies were considered. The financial COI among authors was considered present, absent, or not reported. Additional information on the extracted data regarding Contract Research Organisation (CRO), citation count, journal impact factor (IF), and altmetric attention score (AAS) can be found in Supplementary Table 4. We extracted the main CPG characteristics, including whether they were supported by industry funding, by assessing the funding and acknowledgment section of the CPGs. We explored the CPGs' reference lists and extracted the number of included NMAs and their funding status.

2.4. Outcomes

We assessed the abstracts' conclusions that cited only one treatment as being best and the coherence between the abstracts and the NMAs' results.

Our secondary outcomes were 1/the presence of a registered protocol, 2/the number of included targeted therapies in the NMA over the total number of available targeted therapies at the time of their literature search, 3/the outcomes (efficacy and/or safety) studied in each NMA, 4/the citations per NMA in a 2-year timeframe following the publication, 5/the journal IF at the time of publication, and 6/the NMAs' AAS for the NMAs published in the last 5 years.

2.5. Evaluation of coherence

This evaluation was made in duplicate by independent authors (L.L.C., R.G., and T.B.) following a decisional tree.

The conclusion was divided into propositions, with each sentence corresponding to a proposition. Each proposition that made a statement regarding the relative efficacy or safety of the included intervention was compared to the



Figure 1. Flow chart of the included NMAs and clinical practice guidelines 1A. Flow chart of the included NMAs, 1B. Flow chart of the included CPGs Abbreviations: NMA, network meta-analysis, CPG, clinical practice guideline, IPD, individual patient data, NICE, National Institute for Health and Care Excellence. (For interpretation of the references to color in this figure legend, the reader is referred to the Web version of this article.)

results and noted as being "true" or "false" depending on whether it was supported by the findings or not (Supplementary Figure 1). We examined in the result section the pairwise relative effect estimates with their confidence intervals to determine whether the interventions cited as best were indeed statistically superior to other uncited interventions. If every proposition of the conclusion were true, the conclusion was considered coherent with the results; if not, the conclusion was considered unsupported by the results (Supplementary Figure 2). If the abstract conclusion did not contain any statement regarding the interventions' relative efficacy or safety, the same algorithm was applied to the results section of the abstract.

2.6. Statistical analysis

The characteristics of the NMAs were described; overall and according to the type of funding. Associations between outcomes and funding were analysed by univariate and multivariable logistic regression analyses to compute odds ratios (OR) and the associated 95% confidence intervals (95% CI). Models were adjusted for underlying IMIDs, protocol registration, outcomes of the NMA [efficacy alone, safety alone, or both], exhaustivity of included targeted therapies, and year of publication. These variables were chosen according to already-published articles assessing the association between funding and the presence of spin in the abstract [16-20]. We conducted several sensitivity analyses. Firstly, we excluded studies that did not report their funding, which were at risk of misclassification. We also conducted a sensitivity analysis where we only compared funded NMAs, either from industry or public or academic funding. Lastly, for two additional sensitivity analyses, we excluded NMAs having 1/three authors or less and, subsequently, 2/less than three authors. We considered that a minimum of three authors were needed to conduct a systematic review (two for the duplicated study identification and extraction, and one required for consensus in case of disagreement). These analyses aimed to exclude NMAs that were conducted with the 1/bare minimum number of authors or less, 2/less than the minimum. The threshold for statistical significance was set at 0.05. All analyses were performed on R version 4.3.1.

2.7. Deviation from the protocol

Before starting our review, we excluded from our eligibility criteria cost-effectiveness studies and NMAs that used IPD. We amended the decision tree assessing coherence between the abstracts and the results. Instead of assessing the conclusions in their entirety we assessed the conclusions by proposition to minimize subjective interpretation. For our primary outcomes, the multivariable logistic regressions were added *post-hoc* to account for possibly confounding variables.

3. Results

3.1. NMAs overview

3.1.1. Included studies

Of the 6099 potentially eligible studies, 216 were included (Fig 1A). We failed to retrieve the full text of 13 studies, despite requests to authors (Supplementary Table 6). The full list of the included NMAs is available in Supplementary Table 7. Reference list with reasons for exclusion at full-text stage are reported in Supplementary Table 5.

3.2. CPGs overview

3.2.1. Included CPGs

Of the 8417 potentially eligible reports, 99 CPGs were included (Fig 1B). We failed to retrieve the full texts of two reports despite the authors' solicitation. The full list of included CPGs is available in the Supplementary Table 15.

3.3. Characteristics of the included studies

3.3.1. NMAs' characteristics

The 216 included NMAs were published between 2003 and 2023. All were written in English. A majority studied rheumatoid arthritis (78/216, 36%) followed by psoriasis (52/216, 24%) and IBD (50/216, 23%) (Table 1, Fig 2). 47/216 NMA (22%) registered a protocol, and a clear definition of a primary outcome was available in 40% (86/216) (Table 1). Half of the NMAs assessed both efficacy and safety outcomes (111/216, 52%). Regarding funding, 67/216 (31%) were industry-funded, and among them, 52/67 (78%) declared the assistance of a CRO. Among the nonindustry-funded NMAs, 65/149 (44%) received public or academic funding, 48/149 (32%) did not receive any funding, and 36/149 (24%) did not report whether they received any funding. Characteristics for each study can be found in Supplementary Table 7.

3.3.2. CPGs' characteristics

We identified 32 CPGs on IBD, 11 on psoriatic arthritis, 19 on psoriasis, 22 on rheumatoid arthritis, and 15 on ankylosing spondylitis. The majority of CPGs were national (81/ 99, 82%) and originated from Europe (42/99, 42%). In all, 20/99 (20%) CPGs were supported by either one (9/20, 45%) or several (11/20, 55%) pharmaceutical companies.

3.4. Number of treatments reported as being best

Overall, 173/216 NMAs made a statement regarding the relative efficacy or safety of the included interventions in their abstract's conclusion. Among them, 57/173 (33%) were industry-funded and 116/173 (67%) were not. The proportion of industry-funded NMAs that cited one treatment as being the best was 44% (25/57) compared with 26% (30/116) (Fig 3). Industry-funded NMAs presented a

Table	1.	Main	NMAs'	characteristics	given	the	source	of	fundir	I
-------	----	------	-------	-----------------	-------	-----	--------	----	--------	---

NMA's characteristics	Overall <i>n</i> (%) (total = 216)	Industry-funded NMAs n (%) (total = 67)	Non industry-funded NMAs n (%) (total = 149)	<i>P</i> -value ^a for secondary outcomes
Protocol registration	47 (22)	7 (10)	40 (27)	.01
Financial COI among authors ^b				-
Yes	117 (54)	67 (100)	50 (34)	
No	86 (40)	0	86 (58)	
Not reported	13 (6)	0	13 (9)	
Assistance of a CRO	56 (26)	52 (78)	4 (3)	-
Author from CRO	46 (21)	43 (64)	3 (2)	-
Author from pharmaceutical companies	57 (26)	57 (85)	0	-
Studied population				-
Psoriasis	52 (24)	27 (40)	25 (17)	
Psoriatic arthritis	22 (10)	8 (12)	14 (9)	
IBD	50 (23)	8 (12)	42 (28)	
Ankylosing spondylitis	14 (6)	3 (4)	11 (7)	
Rheumatoid arthritis	78 (36)	21 (31)	57 (38)	
Outcome of interest				<.001
Efficacy alone	85 (39)	43 (64)	42 (28)	
Safety alone	20 (9)	1 (1)	19 (13)	
Both efficacy and safety	111 (51)	23 (34)	88 (59)	
Definition of a primary outcome	86 (40)	9 (12)	77 (52)	<.001
Inclusion of all available targeted therapies	64 (30)	21 (31)	43 (29)	.83
Proportion of included targeted therapies over the number of available targeted therapies	80 (50–71)	85 (67–79)	75 (38–67)	.02
Median number of included targeted therapies, median (IQR)	5 (4-8)	8 (5–10)	5 (3–8)	-
Median number of included studies, median (IQR)	17 (8–33)	20 (10–39)	16 (8–31)	-
Median number of efficacy outcome, median (IQR)	3 (2–4)	3 (3–5)	3 (2-4)	-
Median number of safety outcome, median (IQR)	2 (1–3)	2 (2–3)	2 (1–3)	-

COI, conflict of interest; CRO, contract research organization; IBD, inflammatory bowel disease; IQR, interquartile range; NMA, network metaanalysis.

^a P values from χ^2 -test or Fisher's exact test for categorical variables or Mann–Whitney U test for continuous variables.

^b Financial COI of any type were extracted from the authors' affiliation and the disclosure of conflict of interest section of the publication.

statistically significant risk of citing one treatment as being best before adjustment (OR = 2.24 [1.15–4.39], P = .02), but not after adjustment (aOR = 1.76 [0.81–3.81], P = .15) (variables adjusted for were: underlying IMIDs, protocol registration, NMA's outcomes, exhaustivity of included targeted therapies, and year of publication). These results were consistent across all sensitivity analyses (Supplementary Table 13). For the industry-funded NMAs, the treatment cited as best was always the sponsored intervention (25/25). When several treatments were cited as being best, the sponsored intervention was almost always cited among the best (31/32 [97%]).

3.5. Coherence between the abstracts and the reported results

Half of the abstract's conclusions were labeled as coherent (108/216) and the other half as "unsupported". Agreement between the assessors was fair (percentage agreement = 70%, Cohen's $\kappa = 0.40$). The main reason behind the lack of support was the omission of citing a treatment not statistically different from those listed as the most effective in 93/108 (86%) cases, followed by the statement of a proposition that was true for some but not all the outcomes in 11/108 (10%) cases (Supplementary



Figure 2. Cumulative number of industry and nonindustry-funded NMAs published every year for each IMIDs Abbreviations: NMA, network metaanalysis; IBD, inflammatory bowel disease. (For interpretation of the references to color in this figure legend, the reader is referred to the Web version of this article.)

Table 12). NMAs with industry-funding presented a nonsignificant risk of presenting unsupported conclusions before (OR = 1.61 [0.90-2.89], P = .11) and after adjustment (aOR = 1.40 [0.71-2.78], P = .33). These results were consistent in all sensitivity analyses (Supplementary Table 13).

3.6. Comparison between industry-funded and nonindustry-funded NMAs

We found that industry-funded NMAs were less likely to register a protocol (P = .01), less likely to define primary outcomes (P < .001), less likely to evaluate both efficacy and safety outcomes (P < .001), and more likely to evaluate solely efficacy outcomes compared to nonindustry-

funded NMAs (P < .001). Industry-funded NMAs included a higher proportion of the available treatments; however, we found no difference between the sources of funding regarding the proportion of NMAs that included all the available targeted therapies (Table 1).

3.7. Effect of funding on NMAs dissemination and use in CPGs

We did not find any difference in NMAs' dissemination between industry-funded and nonindustry-funded NMAs in terms of the median number of citations, median AAS, and median journal IF (Supplementary Table 18).

Regarding their use in CPGs, 59/99 (60%) cited at least 1 NMA, with a median number of cited NMAs of 2



Figure 3. Coherence of the abstract conclusion given the number of treatments cited as best and the source of funding Abbreviation: NMA, network meta-analysis. (For interpretation of the references to color in this figure legend, the reader is referred to the Web version of this article.)

(IQR = 1-4). All the NMAs cited in the CPGs were included in our review of NMAs, except for 7 that did not fit our inclusion criteria. Among the CPGs that referenced NMAs, 6/59 (10%) cited industry-funded NMAs only, 36/59 (61%) cited nonindustry-funded NMAs only, and 17/59 (29%) cited both. We did not find any association between the funding of CPGs by the industry and the presence of NMAs in the reference list (P = .60) or the citation of industry-funded NMAs (P = .71).

4. Discussion

Given the expansion of therapeutic options, NMA stands out as a valuable tool for clinical decision-makers and CPG developers. This method, however, may be an opportunity for pharmaceutical companies to use it as a marketing tool to promote their drugs. In our study, a third of the included NMAs were industry-funded.

Industry-funded NMAs had a 40% increase in odds to have conclusions unsupported by the results and a 76% increase in odds to cite only one treatment as being best in their conclusions. However, there was uncertainty about these associations. Still, almost all favored their own treatments, either as a unique option or as one of the best options. The main reason behind unsupported conclusions was the omission of citing a treatment not statistically different from those listed as the most effective, which was more often the case when only one treatment was cited as the best. This observation could be considered a form of spin for NMAs. Nevertheless, even though previous studies have shown that industry funding had an influence on the results and conclusion of systematic reviews and meta-analyses, no statistically significant association was found between industry funding and the occurrence of spin [13,17–19].

Beyond that, we found that no matter the type of funding, a majority of the NMAs failed to meet important methodological requirements. Most of them did not register a protocol, and less than half defined a primary outcome. Moreover, 40% did not evaluate the safety of the included interventions. While found no matter the type of funding, these flaws were more likely to be observed in industryfunded NMAs. Additionally, less than a third of the NMAs included all available targeted therapies at the time of their review. Even though 10% of the NMAs in our overview focused on subpopulation of interest for which the inclusion of all available targeted therapies may have been unnecessary or unreasonable, most of them dismissed valuable information from important interventions.

Our results agreed with the ones presented by Veroniki et al. [12]. One methodological difference is that we assessed the coherence of the reported results. Ultimately, the prevalence of unsupported conclusions was high irrespective of funding, with a bias toward sponsored drugs for industry-funded NMAs. However, the absence of protocol or predefined primary outcomes did not allow us to exclude cases of selective reporting or publication bias. As previously pointed out, the industry often commissions many NMAs through CRO, with most of them not being registered and published after analyses [21]. This raises some concerns that these publications are biased and open to *post-hoc* manipulation of results.

Lastly, only 60% of the included CPGs relied on NMAs, with 39% of them citing industry-funded NMAs, either alone (14%), or with nonindustry-funded NMAs (25%). The proportion of CPGs that included NMAs remained stable over time, despite the more recent incentives for using them to guide recommendations [22,23]. Based on our results, it is possible that guideline developers are reluctant to include industry-funded NMAs and/or biased or poor-quality NMAs, even if this would mean ignoring useful data. A future study will further aim to assess CPGs' financial ties with the industry and the impact of the NMAs on their recommendation.

Our study has several limitations. Firstly, a quarter of the nonindustry-funded NMAs did not report their funding status, which may have led to some misclassification. Additionally, financial COI among authors may have been underestimated as they were extracted from the publications self-reported disclosures [24]. Secondly, the review authors were not blinded to the funding status when assessing the abstracts' coherence. To limit subjective interpretation, the authors followed a decision tree independently and in duplicate. The agreement between assessors was classified as fair, which could be mainly explained by one of the assessors missing nonsignificant differences for some comparisons. All cases of disagreement were resolved through discussions. Thirdly, we focused our attention on the abstracts rather than on the discussion or the main conclusion. We assumed that the abstract would be more impactful on readers. Lastly, we included in our logistic models relevant variables previously acknowledged in the literature [18-20]. Still, there may be some concern regarding the mediation role of some variables, questioning their classification as true confounders. While the inclusion of these potentially mediating variables could theoretically introduce bias, their incorporation tended to shift estimations toward the null.

5. Conclusion

Although there was no evidence that industry-funded NMAs were more likely to have unsupported conclusions or to cite only one treatment as best in their conclusions, almost all favored their own treatments either as a unique option or as one of the best options. Even though the use of NMA was modest in the included CPGs, over a third of those who did used industry-funded NMAs.

Patient and public involvement

There was no patient or public involvement during the conduct of this study.

Ethics approval

No ethics approval was required for this study.

Transparency declaration

The lead author affirms that this manuscript is an honest, accurate, and transparent account of the study being reported; that no important aspects of the study have been omitted; and that any discrepancies from the study as planned (and, if relevant, registered) have been explained.

CRediT authorship contribution statement

Robin Guelimi: Writing – review & editing, Writing – original draft, Visualization, Validation, Project administration, Methodology, Funding acquisition, Formal analysis, Data curation, Conceptualization. Sivem Afach: Writing - review & editing, Writing - original draft, Visualization, Validation, Methodology, Conceptualization. Thomas Bettuzzi: Writing - review & editing, Writing - original draft, Visualization, Validation, Data curation. Antoine Meyer: Writing - review & editing, Writing - original draft, Visualization, Validation, Methodology, Data curation. Guillaume Padern: Writing - review & editing, Writing - original draft, Visualization, Validation, Methodology, Data curation. Zenas Yiu: Writing - review & editing, Writing - original draft, Visualization, Validation, Methodology, Conceptualization. Florian Naudet: Writing - review & editing, Writing - original draft, Visualization, Validation, Methodology, Conceptualization. Emilie Sbidian: Writing – review & editing, Writing – original draft, Visualization. Validation. Supervision, Methodology, Conceptualization. Laurence Le-Cleach: Writing - review & editing, Writing - original draft, Visualization, Validation, Supervision, Project administration, Methodology, Conceptualization.

Data availability

Data and code underlying this article are available on Open Science Framework (osf.io/wgydt).

Declaration of competing interest

Robin Guelimi reports financial support was provided by Public Assistance Hospitals Paris. and the Association for

Interdisciplinary Meta-Research and Open Science. LLC, ES, and SA are co-authors of this work and authors of several studies that are included in the submitted overview. L.L.C., E.S., and S.A. are authors of several studies that are included in this overview. L.L.C. took part in the selection process, but none of them were involved in the extraction of data or in the assessment of the quality of a work they took a part in. L.L.C. and E.S. received a grant from the French Ministry of Health (PHRC-N AOM22145). L.L.C. is the co-ordinating editor of the Cochrane Skin. F.N. received funding from the French National Research Agency (ANR-17-CE36-0010), the French ministry of health and the French ministry of research. He is a work package leader in the OSIRIS project (Open Science to Increase Reproducibility in Science). The OSIRIS project has received funding from the European Union's Horizon Europe research and innovation programme under the grant agreement No. 101094725. He is a work package leader for the doctoral network MSCA-DN SHARE-CTD (HORI-ZON-MSCA-2022-DN-01 101120360) funded by the EU. YZZN received a grant from the NIHR Manchester Biomedical Research Centre (NIHR203308). R.G., A.M., T.B., and G.P. had no conflict of interest to declare. If there are other authors, they declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

The authors would like to thank the « Assistance Publique des Hôpitaux de Paris » as well as the Association for Interdisciplinary Meta-Research and Open Science (AI-MOS) for their financial support of this research. The authors would also like to thank Altmetrics for lending access to their database. This research was supported by the NIHR Manchester Biomedical Research Centre (NIHR203308). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care.

Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jclinepi.2024.111411.

References

- Davies C. Dermatology drives autoimmune market growth [online]. 2022. Available at: https://www.iqvia.com/-/media/iqvia/pdfs/ library/articles/iqvia-forecast-link-autoimmune-market-article.pdf. Accessed December 8, 2023.
- [2] Lindsley CW. New 2016 data and statistics for global pharmaceutical products and projections through 2017. ACS Chem Neurosci 2017; 8(8):1635–6.

- [3] McInnes IB, Gravallese EM. Immune-mediated inflammatory disease therapeutics: past, present and future. Nat Rev Immunol 2021;21: 680-6.
- [4] Institute of Medicine (US) Committee on Standards for Developing Trustworthy Clinical Practice Guidelines. Available at:. In: Graham R, Mancher M, Miller Wolman D, et al, editors. Clinical Practice Guidelines We Can Trust. Washington (DC): National Academies Press (US); 2011. https://www.ncbi.nlm.nih.gov/books/ NBK209539/doi:10.17226/13058. Accessed February 15, 2024.
- [5] Kanters S, Ford N, Druyts E, Thorlund K, Mills EJ, Bansback N. Use of network meta-analysis in clinical guidelines. Bull World Health Organ 2016;94(10):782–4.
- [6] Leucht S, Chaimani A, Cipriani AS, et al. Network meta-analyses should be the highest level of evidence in treatment guidelines. Eur Arch Psychiatr Clin Neurosci 2016;266:477–80.
- [7] Afach S, Evrenoglou T, Oubaya N, Le Cleach L, Sbidian E. Most randomized controlled trials for psoriasis used placebo comparators despite the availability of effective treatments. J Clin Epidemiol 2021;133:72–9.
- [8] Singh S, Murad MH, Fumery M, et al. Comparative efficacy and safety of biologic therapies for moderate-to-severe Crohn's disease: a systematic review and network meta-analysis. Lancet Gastroenterol Hepatol 2021;6(12):1002–14.
- [9] Chaimani A, Caldwell DM, Li T, Higgins JPT, Salanti G. Chapter 11: undertaking network meta-analyses. In: Higgins JPT, Thomas J, Chandler J, Cumpston M, Li T, Page MJ, et al, editors. Cochrane Handbook for Systematic Reviews of Interventions version 6.4. Cochrane; 2023. (updated August 2023) www.training.cochrane.org/ handbook. Accessed June 26, 2024.
- [10] Naudet F, Schuit E, Ioannidis JPA. Overlapping network metaanalyses on the same topic: survey of published studies. Int J Epidemiol 2017;46:1999–2008.
- [11] Leonardi-Bee J, Drucker AM. Systemic treatments for psoriasis: not another network meta-analysis. Br J Dermatol 2022;187:3–4.
- [12] Veroniki AA, Wong EKC, Lunny C, et al. Does type of funding affect reporting in network meta-analysis? A scoping review of network meta-analyses. Syst Rev 2023;12(1):81.

- [13] Guelimi R, Afach S, Régnaux JP, et al. Overlapping network metaanalyses on psoriasis systemic treatments, an overview: quantity does not make quality. Br J Dermatol 2022;187:29–41.
- [14] Gates M, Gates A, Pieper D, et al. Reporting guideline for overviews of reviews of healthcare interventions: development of the PRIOR statement. BMJ 2022;378:e070849.
- [15] Yen H, Huang CH, Huang IH, et al. Systematic review and critical appraisal of psoriasis clinical practice guidelines: a Global Guidelines in Dermatology Mapping Project (GUIDEMAP). Br J Dermatol 2022;187:178–87.
- [16] Yavchitz A, Ravaud P, Altman DG, et al. A new classification of spin in systematic reviews and meta-analyses was developed and ranked according to the severity. J Clin Epidemiol 2016;75:56–65.
- [17] Chiu K, Grundy Q, Bero L. 'Spin' in published biomedical literature: a methodological systematic review. PLoS Biol 2017;15(9): e2002173.
- [18] Wise A, Mannem D, Arthur W, et al. Spin within systematic review abstracts on antiplatelet therapies after acute coronary syndrome: a cross-sectional study. BMJ Open 2022;12:e049421.
- [19] Lin V, Patel R, Wirtz A, et al. Evaluation of spin in the abstracts of systematic reviews and meta-analyses of atopic dermatitis treatments and interventions. Dermatology 2021;237(4):496–505.
- [20] Faulkner JJ, Polson C, Dodd AH, et al. Evaluation of spin in the abstracts of systematic reviews and meta-analyses focused on the treatment of obesity. Obesity 2021;29(8):1285–93.
- [21] Schuit E, Ioannidis JP. Network meta-analyses performed by contracting companies and commissioned by industry. Syst Rev 2016;5:198.
- [22] Ceriello A, Rodbard HW, Battelino T, et al. Data from network metaanalyses can inform clinical practice guidelines and decision-making in diabetes management: perspectives of the taskforce of the guideline workshop. Cardiovasc Diabetol 2023;22:277.
- [23] Antoniou SA, Mavridis D, Tsokani S, et al. Network meta-analysis as a tool in clinical practice guidelines. Surg Endosc 2023;37:1–4.
- [24] Taheri C, Kirubarajan A, Li X, et al. Discrepancies in self-reported financial conflicts of interest disclosures by physicians: a systematic review. BMJ Open 2021;11:e045306.